Author’s response to reviews

Title: Trends and correlates of driving under the influence of alcohol among different types of adult substance users in the United States: A national survey study

Authors:
jiyeun park (pjy0225@gmail.com)
Li-Tzy Wu (litzy.wu@duke.edu)

Version: 2 Date: 29 Mar 2019

Author’s response to reviews:

We have attached a copy of authors' response to the reviewers' comments as a supplementary document.